Author’s response to reviews

Title: Reliability and Validity of a Novel Kinect-based Software Program for Measuring Posture, Balance and Side-bending

Authors:

Wilhelmus Grooten (wim.grooten@ki.se)
Lisa Sandberg (lisamargaretasandberg@gmail.com)
John Ressman (john@medicinskpt.se)
Nicolas Diamantoglou (nicodiamant@gmail.com)
Elin Johansson (Elin.Johansson@ki.se)
Eva Rasmussen-Barr (eva.rasmussen-barr@rehabtjanst.se)

Version: 1 Date: 23 Oct 2017

Author’s response to reviews:

Thank you for your email.

I believe that I have made all the changes that were necessary. I have marked the changes with yellow in the manuscript.

On behalf of all authors,

Best wishes

Wim Grooten